Author's response to reviews

Title: Growth of a cohort of very low birth weight infants in Johannesburg, South Africa

Authors:

Cheryl A Mackay (cheryl.mackay@wits.ac.za)
Daynia E Ballot (daynia.ballot@wits.ac.za)
Peter A Cooper (peter.cooper@wits.ac.za)

Version: 7 Date: 20 May 2011

Author's response to reviews:

To The Editorial Team,

Further revisions to the manuscript titled "Growth of a cohort of very low birth weight infants in Johannesburg, South Africa" have been made, including changes to referencing, numbering of sections and removal of all vertical lines in the tables. I have reviewed the figures and found that to the best of my knowledge they are formatted according to the BMC guideline. Please let me know if there is anything specific that needs to be changed with them. The manuscript has been resubmitted for your attention.

Kind regards,
Cheryl Mackay